Author's response to reviews

Title: Impact of intraoperative fluid administration on outcome in patients undergoing robotic-assisted laparoscopic prostatectomy - a retrospective analysis

Authors:

Tobias Piegeler (tobias.piegeler@uzh.ch)
Pamela Dreessen (pamela.dreessen@usz.ch)
Sereina Graber (sereina.graber@gmx.ch)
Sarah R Haile (sarah.haile@uzh.ch)
Daniel M Schmid (daniel.max.schmid@usz.ch)
Beatrice Beck-Schimmer (beatrice.beckschimmer@uzh.ch)

Version: 4  Date: 12 July 2014

Author's response to reviews: see over
Dear Editor,

Thank you very much for accepting our manuscript entitled ‘Impact of intraoperative fluid administration on complications and length of hospitalisation in patients undergoing robotic-assisted laparoscopic prostatectomy – a retrospective analysis’ for publication in BMC Anesthesiology.

As requested, we have now included the name and address of our local ethics committee in our manuscript.

All authors have read the revised manuscript, attest to the validity and legitimacy of the data and its interpretation, and agree to its re-submission to BMC Anesthesiology. All authors declare they have no competing interests.

Sincerely,

Tobias Piegeler, MD

Tobias Piegeler, MD
University Hospital Zurich
Institute of Anesthesiology
Rämistrasse 100
CH-8091 Zürich
Switzerland

Phone Secretary +41 44 255 26 96
Fax +41 44 255 44 09
tobias.piegeler@uzh.ch

Zurich, 07/12/2014
Response to Editor's comments

Thank you for including an ethics statement in your manuscript. However, we would ask you to include the full name of the committee that approved your study in this statement.

*We have now included the name and address of our local ethics committee in our manuscript.*